Author's response to reviews

Title: Closing the osteoporosis care gap - Increased osteoporosis awareness among geriatrics and rehabilitation teams.

Authors:

Derek A Haaland (haaland@mcmaster.ca)
Dana R Cohen (dana.cohen@sunnybrook.ca)
Courtney C Kennedy (kennedyc@hhsc.ca)
Nadir A Khalidi (khalidn@mcmaster.ca)
Jonathon D Adachi (jd.adachi@sympatico.ca)
Alexandra Papaioannou (papaioannou@hhsc.ca)

Version: 4 Date: 22 June 2009

Author's response to reviews:

June 22nd, 2009

Dear Committee,

This latest submission addresses the suggestions made in an email to Dr. Papaioannou dated June 17th, 2009. The only minor issue that hasn't been addressed is with regards to "figure cropping". The figure looks to be reasonably cropped, but please let me know if a further change is required by us here. One additional reference (21) with respect to "academic detailing" has been added, and a number of small typographical issues have been addressed.

We look forward to your reply. Again, please let us know of any further concerns or suggestions.

Sincerely,

Derek Haaland
on behalf of the study coauthors